#### **ORIGINAL ARTICLE**



# An experimental test of changed personality in butterflies from anthropogenic landscapes

Aurélien Kaiser<sup>1</sup> · Thomas Merckx<sup>1,2</sup> · Hans Van Dyck<sup>1</sup>

Received: 25 February 2020 / Revised: 13 May 2020 / Accepted: 8 June 2020 / Published online: 17 June 2020 © Springer-Verlag GmbH Germany, part of Springer Nature 2020

#### Abstract

During the last century, the human footprint on natural ecosystems has increased strongly and human-altered habitats such as urban and agricultural areas have extended globally. Despite their negative impacts on biodiversity, these habitats offer unique opportunities to study how native species respond to novel environmental conditions. Here, we studied phenotypic divergence associated with colonization of human-altered habitats in the Speckled wood (*Pararge aegeria*). We reared butterflies of woodland, urban and agricultural origins under common garden conditions and we measured boldness and activity at the adult stage. Both behavioural traits were repeatable at the individual level (i.e. personality traits), but we found weak evidence for ecotype-related differences in mean boldness and activity. In line with urban areas being stressful habitats, we found that boldness and activity traits correlate in urban butterflies, while we found no such syndrome in woodland and agricultural butterflies. Our results show that urbanization can alter some aspects of personality in an insect species, but they do not support the prediction that anthropogenic habitats favour boldness.

## Significance statement

Human activities such as urbanization and intensive agriculture strongly alter terrestrial ecosystems and they are among the most significant threats to biodiversity. To tolerate human-dominated landscapes, many vertebrate species show behavioural shifts towards bold personalities, but similar responses remain rather overlooked in invertebrate taxa. Here, we studied the progeny of woodland, agricultural and urban Speckled woods reared under common garden conditions and we assessed their personality. We found little evidence for differences in personality traits among landscape types, but the behavioural syndrome linking boldness and activity was detected only in urban butterflies. This shows that urbanization can indeed shape some aspects of personality in invertebrates.

**Keywords** Animal personality  $\cdot$  Behavioural syndrome  $\cdot$  Ecotypic differentiation  $\cdot$  Human-induced rapid environmental change  $\cdot$  Urbanization

Communicated by J. C Choe

**Electronic supplementary material** The online version of this article (https://doi.org/10.1007/s00265-020-02871-8) contains supplementary material, which is available to authorized users.

# Introduction

Human activities have become a major evolutionary driver for other species. Through behaviours such as selective hunting, large-scale harvesting, species translocation and landscape modifications, *Homo sapiens* has proven to be an effective selective pressure affecting the phenotype of a wide range of wild species (Allendorf and Hard 2009; Sullivan et al. 2017). The process of urbanization, i.e. the replacement of (semi)natural vegetation (including agricultural land uses) by buildings and other impermeable surfaces and infrastructure, involves some of the most extreme forms of human-caused landscape changes. It results in increased fragmentation of natural habitats and in altered biophysical processes (Parris 2016). Consequently, urban landscapes typically have

Aurélien Kaiser kaiser.aurelien@gmail.com

<sup>&</sup>lt;sup>1</sup> Behavioural Ecology & Conservation Group, Earth & Life Institute, UCLouvain, Louvain-la-Neuve, Belgium

<sup>&</sup>lt;sup>2</sup> Department of Ecology and Genetics, University of Oulu, Oulu, Finland

reduced species richness compared to surrounding natural habitats (McKinney 2008; Concepción et al. 2015; Ramírez-Restrepo and MacGregor-Fors 2017; Merckx et al. 2018a, b; Piano et al. 2020). Yet, in addition to human commensals (e.g. feral pigeons, rats), cities typically harbour native species that persisted locally despite severe landscape modifications and species that have colonized urban areas more recently from adjacent (semi-) natural habitats. In Europe, one of the most iconic examples of such colonization is the European blackbird (*Turdus merula*), which was originally a forest specialist, but started breeding in urban gardens and parks about 200 years ago (Luniak et al. 1990). The urban ecotype then spread over most of the species range in Western and Central Europe following independent colonization events (Evans et al. 2009).

Species confronted with novel habitats face numerous challenges in the form of changes to the physical and/or biotic environment (Hobbs et al. 2009). Consequently, successful colonization of anthropogenic landscapes is likely to require adjustments at multiple levels and urban animals often differ from their rural conspecifics in terms of behaviour (reviewed in Sol et al. 2013), morphology (Winchell et al. 2016; Merckx et al. 2018a; Kern and Langerhans 2018) and life-history or physiological traits (Sprau et al. 2017; Tüzün et al. 2017b; Sepp et al. 2018; Diamond et al. 2018). Both genetic differentiation and phenotypic plasticity may contribute to such landscape-related differences, although the relative contribution of each mechanism is often difficult to assess. Nonetheless, we now know that some of these differences indeed have at least partly a genetic basis (Alberti et al. 2017).

Animal personality refers to consistent behavioural differences among individuals (Réale et al. 2007). Within a species, individuals may consistently differ in behavioural traits such as boldness, activity, exploration, aggressiveness or sociability, which are the major personality axes (Réale et al. 2007). Moreover, personality traits tend to be grouped into behavioural syndromes, whereby the average phenotype of individuals in one context is correlated with the average phenotype of the same individuals in another context (i.e. between-individual correlation - Dingemanse et al. 2012). For instance, Great tits (Parus major) that explore a new environment more rapidly are also more aggressive towards conspecifics (Verbeek et al. 1996). Personality traits and behavioural syndromes are relevant in the context of human-induced rapid environmental changes, as they have widereaching implications for ecological and evolutionary processes (Sih et al. 2004; Sol et al. 2013). Several studies also reported shifts in personality traits linked to urbanization. For instance, a study across multiple European cities showed that wild urban Great tits displayed more distress calling and higher pecking rates (i.e. an index of handling aggression) than their rural conspecifics (Senar et al. 2017). Overall, urban vertebrates show personality traits associated with a more proactive life-style (Koolhaas et al. 1999): urban animals tend to be more explorative, bolder and more active than their rural conspecifics. Such a shift towards a

more proactive style is supposed to be adaptive as it would reduce fear reactions towards novel objects and it would facilitate the acquisition of new food resources (e.g. Sol et al. 2013; Senar et al. 2017). Also, urbanization was found to uncouple personality traits, with behavioural syndrome breakdowns observed in some bird species (e.g. Evans et al. 2010; Bókony et al. 2012).

Comparatively few studies investigated whether similar shifts also occur in invertebrate taxa (Schuett et al. 2018). In this regard, the Speckled wood (Pararge aegeria L.) is an interesting study system for exploring changes in personality traits linked to anthropogenic activities. In NW-Europe, the species inhabits large forest patches, but also small woodlots and hedgerows in agricultural areas (e.g. Merckx and Van Dyck 2005, 2007) and forested gardens and parks in cities (e.g. Bergerot et al. 2012). Together, the three landscape types represent a gradient from relatively natural landscapes (i.e. woodlands) to moderately and strongly human-altered landscapes (i.e. agricultural and urban areas, respectively). Moreover, several studies confirmed that the Speckled wood exhibits consistent behavioural differences (i.e. the existence of personality Ducatez et al. 2014; Kaiser et al. 2018, 2019a, b). In a previous study, we used a field reciprocal transplant approach including all three ecotypes of the Speckled wood and we detected higher boldness levels in agricultural males (Kaiser et al. 2019a). Here, we tested for evolved personality differences (i.e. boldness and activity) between the three ecotypes using butterflies reared under common garden conditions using families collected from 10 regions in central and northern Belgium. This experimental design aims to provide one of the most comprehensive studies so far of human-induced shifts in personality traits and behavioural syndromes in an invertebrate species. We expect butterflies from anthropogenic (i.e. agricultural and urban) landscapes to be bolder and more active, like what is observed in vertebrate taxa. Additionally, there is evidence that adverse conditions, including high predation pressure (Bell 2005; Dingemanse et al. 2007; Urszán et al. 2015) or high pesticide levels (Tüzün et al. 2017a, but see Royauté et al. 2015), can impact the structure of behavioural syndromes. Because warmer and dryer climatic conditions in agricultural and urban landscapes (e.g. Serruys and Van Dyck 2014; Kaiser et al. 2016; Merckx et al. 2018b) are expected to provide stressful conditions for a drought-sensitive species such as the Speckled wood (e.g. Talloen et al. 2004; Oliver et al. 2015), we therefore expect tightened behavioural syndromes in butterflies from anthropogenic landscapes.

## Methods

## Study species and sampling sites

The Speckled wood (*P. aegeria* L.) is a common multivoltine butterfly species that occurs over most of Europe (Settele et al. 2008). Larvae feed on the leaves of various grass species

(Shreeve, 1986). In June and July 2015, we collected gravid females from ten regions in central Belgium. In each region, we selected one woodland, one urban and one agricultural site (see Fig. S1). Woodland sites consisted of continuous forests typically larger than 100 ha; agricultural sites of systems of small woodlots and hedgerows surrounded by arable fields and pastures; urban sites of wooded parks and woodlots surrounded by buildings and roads (Kaiser et al. 2019a). We captured one or two females at each site, which in total resulted in 14 woodland, 12 urban and 15 agricultural families (see Table S1 for capture details). Collected females were transported to the laboratory for oviposition in individual cages on the grass *Poa pratensis*, while they could feed ad libitum on honey-soaked (10% solution) cotton pads.

## **Butterfly rearing**

The offspring (i.e. F1-generation) of the wild-caught females were reared on potted *P. pratensis* plants (grown on a standardized soil mixture under identical light and temperature conditions) in a climate room (photoperiod: L:D 16 h:8 h; day temperature: 25 °C; night temperature: 16 °C). Each host plant contained four full-sib caterpillars and was enclosed in nylon netting. Pots were checked on a regular basis for pupating larvae. Pupae were removed from their plant and placed individually in labelled plastic cups. We checked all cups twice a day for emergence. We used a unique alphanumeric code to identify all individuals. The code was non-informative regarding to the ecotype and it ensured that the observer was blind to the origin of the butterfly during the behavioural tests.

## **Behavioural tests**

We subjected adult offspring to two behavioural assessments: (i) a boldness test, and (ii) an activity test conducted in a novel environment. We tested 50 woodland (20 males; 30 females), 53 urban (22 males; 31 females) and 47 agricultural (20 males; 27 females) butterflies. The number of tested individuals per family is presented in Table S2. Assessments started on the day following emergence, i.e. on day 1, and we retested some individuals on day 2, 3 and 4 (see below). Butterflies had no access to food, but they all had access to a water-soaked cotton pad at the end of each day. A single observer (AK) conducted all behavioural observations. The exact procedure for assessing boldness and activity in the study system has been described in extenso elsewhere (Kaiser et al. 2019a, b), so we only briefly describe the methodology hereafter.

## **Boldness**

We placed the butterfly (with closed wings) in the centre of a semi-transparent glassine envelope and we counted the number of struggles for one minute. Struggles are defined as series of leg, head and/or wing movements, interrupted from other such series by pauses of inactivity. Pauses typically lasted at least a couple of seconds, which allowed a clear distinction between successive struggle bouts. Note that the focus was on the number of bouts and we did not consider their intensity (e.g. duration or number of leg movements within a bout).

We maintained constant light conditions and a room temperature of 25 °C during the test. All butterflies but one were submitted four times to this test, with one day elapsing between two successive trials. Butterflies were then weighed using a microbalance (Ohaus Explorer; accuracy:  $\pm 0.1$  mg). After weighing, butterflies returned to their individual plastic cup until the activity test.

## Activity

The activity test took place in an empty plastic greenhouse tunnel (length  $\times$  width  $\times$  height:  $12 \times 4 \times 2$  m; installed in a larger glass greenhouse) whose floor was taped to delineate two rows of eight rectangles (each  $1.5 \times 2$  m). Each butterfly was released individually at one extremity of the tunnel and it could move freely for four minutes while the observer recorded the number of transitions between rectangles (used as a proxy for activity). Here, we did not consider vertical movements as these generally occurred in a gradual way during transitioning between rectangles. Ambient temperature was on average  $30.0 \pm 5.8$  °C (mean  $\pm$  SD). Hence, this provided optimal thermal conditions for flight (Shreeve 1984; Van Dyck and Matthysen 1998). Due to time constraints, individuals vary in the number of times they were tested for activity. 72 butterflies were tested four times, 33 were tested three times, 8 were tested twice, 8 were tested once and 29 butterflies could not be tested at all. The number of tested individuals per ecotype for the activity test is presented in Table S3. Again, one day elapsed between two successive trials.

#### **Statistical analysis**

All statistical analyses were performed with R 3.5.1 (R Core Team 2020).

## Effects of fixed variables on mean behaviour

We used linear mixed models (package *lme4*) to test for differences in behaviour among ecotypes and sexes, and to unravel dynamics of these traits relative to the testing sequence. Prior to the analyses, we applied a square-root transformation to boldness and activity to achieve normality of the residuals. Boldness and activity were the response variables, while ecotype (i.e. woodland, urban or agricultural), sex (i.e. male or female), sequence (as a categorical variable) were included as explanatory variables. We initially included the ecotype × sex and ecotype × sequence interactions, but they were never significant ( $P \ge 0.1$ ) and we thus removed them from the final models. Region of origin, family ID and individual ID were included as random factors (i.e. random intercepts) in all models. Body mass was added as a covariate for both behavioural traits. For activity, we added temperature in the greenhouse tunnel (in °C) and solar irradiance (in W/m<sup>2</sup>) during the test as additional covariates. Continuous variables (i.e. covariates) were scaled prior to the analyses.

## Repeatability of behavioural traits

We calculated individual repeatability to validate that consistent individual differences (i.e. personality) in boldness and activity occurred in our study system. Repeatability is the fraction of the total phenotypic variance that can be attributed to between-individual differences (Nakagawa and Schielzeth 2010). From the mixed models presented above, we simulated the posterior distribution of the random terms based on 2000 simulations using the *sim* function from the arm package. We calculated the adjusted repeatability of boldness and activity based on Nakagawa and Schielzeth (2010). Repeatability was considered significant when 95% credible intervals did not overlap zero.

#### **Correlations among behavioural traits**

To test for correlations among behavioural traits, we fitted ecotype-specific bivariate mixed models with the MCMCglmm package (Hadfield 2010). Models contained boldness and activity (both square-root transformed) as response variables, sex, sequence and body mass as fixed effects, and region of origin, family ID and individual ID as random effects. We used a Gaussian distribution and a noninformative inverse-Wishart prior. We used 300,000 iterations, from which we discarded the first 30,000 (burn-in), while using a thinning interval of 100. This resulted in low autocorrelation. We performed three runs to confirm robustness of the model outputs. We report within- and betweenindividual correlations as they both contribute to phenotypic correlations. The former represents the correlation between an individual's change in one trait between t and t + 1 and the change in another trait over the same period (Dingemanse and Dochtermann 2013). The latter represents the correlation between individual mean values of two traits and it is thus a measure of behavioural syndromes sensu stricto (Dingemanse et al. 2012). Within- and between-individual correlations were estimated following equations in Dingemanse and Dochtermann (2013). Correlations whose 95% credible intervals do not overlap with zero were considered significant.

#### Results

#### **Boldness and activity**

Boldness showed a moderate repeatability (Table 1) and this behavioural trait was influenced by testing sequence  $(F_{3,467.6} = 12.71; P < 0.0001)$ : boldness increased from the first to the second day and then remained stable until the last test on day 4 (Fig. 1). The overall effect of ecotype was not statistically significant ( $F_{2,25.5} = 3.16; P = 0.059$ ), but post-hoc comparisons revealed that agricultural butterflies had lower boldness scores on average than urban butterflies (P = 0.042after Bonferroni adjustment for multiple comparisons) (Fig. 1). Other post-hoc comparisons were non-significant (Posthoc test: P > 0.20 after Bonferroni adjustment). We did not detect any effect of sex ( $F_{1,362.2} = 1.33; P = 0.249$ ), nor of body mass ( $F_{1,508.9} = 0.58; P = 0.448$ ) on this trait.

Activity was also repeatable, although the repeatability was lower than for boldness (Table 1). Contrary to boldness, none of the fixed effects considered had a significant effect on activity (all *P*-values  $\geq 0.09$ ).

## **Correlations among traits**

A significant among-individual correlation between boldness and activity was present in the urban ecotype only (Table 2). For all ecotypes, within-individual correlation between boldness and activity was low and non-significant.

# Discussion

As the human pressure on natural systems increases, many species are confronted with novel environmental conditions. Using Speckled woods of woodland, agricultural and urban population origins reared under common garden conditions, we show weak differentiation in mean boldness (but not in mean activity) among populations from the three landscape types. Yet, the behavioural syndrome (i.e. among-individual correlation) linking these two traits was only detected in urban-origin butterflies. This provides an example of a phenotypic change associated with human-altered habitats whose footprint is forecasted to increase globally over the next decades.

Evidence for personality differences among rural and urban populations is accumulating in animals (e.g. Miranda et al. 2013; Lapiedra et al. 2017; Charmantier et al. 2017; Senar et al. 2017; Schuett et al. 2018; Baxter-Gilbert et al. 2019). However, most studies on urban-rural behavioural differences focused on wild individuals. It is therefore unclear whether these differences reflect a genetic basis or arise due to phenotypic plasticity (but see Tüzün et al. 2017a). Anthropogenic landscapes typically have altered plant and animal

 Table 1
 Variance components associated with each of the random effects included in the analyses of boldness and activity, with 95% confidence intervals. We also show the individual repeatability for each trait (r Individual, with 95% credible intervals)

	$\sigma^2$ Individual	$\sigma^2_{Family}$	$\sigma^2_{Region}$	$\sigma^2_{\text{Residual}}$	r Individual
Boldness	0.846 [0.692; 1.027]	0.105 [0.059; 0.164]	0.000 [0.000; 0.000]	0.896 [0.816; 1.024]	0.442 [0.400; 0.503]
Activity	1.368 [1.024; 0.970]	0.609 [0.372; 0.970]	0.010 [0.003; 0.030]	8.769 [7.449; 9.806]	0.133 [0.102; 0.159]

communities and often show increased disturbance due to human activities (e.g. recreational activities or agricultural management practices). Just as observed in vertebrate species, traits associated with a proactive style may facilitate the exploitation of new types of resources (e.g. new nectar sources or prey items) and increase the resilience to disturbance in invertebrate taxa too. We may thus expect that high boldness and activity levels would be adaptive in urban (and agricultural) environments and that this would translate into evolved (i.e. genetically based) differences, even though responses to urbanization are likely to depend ultimately on species-specific attributes (e.g. sensory ecology, ecological niche). Contrary to our expectations, we found little evidence for ecotype-related differences in mean boldness and not at all for activity in our Speckled wood samples reared under a common garden setting. For example, woodland and urban butterflies show no difference in mean levels of personality traits even though they originate from the most contrasting landscape types in our system. This result echoes our previous findings (Kaiser et



Fig. 1 Effects of sequence and ecotype on boldness. Points show the expected mean  $\pm$  95% confidence intervals based on the model output

al. 2018). While it remains to be tested whether wild Speckled wood individuals show differences in mean levels of personality traits among ecotypes, this result implies that such differences would be mainly caused by behavioural plasticity. We may even speculate on the developmental stage at which behavioural differences among urban and woodland butterflies would arise. Indeed, a reciprocal experiment conducted under field conditions showed that conditions (i.e. different landscape types) experienced during the larval and early pupal stages had no effect on mean levels of personality traits in the Speckled wood (Kaiser et al. 2019a). Thus, conditions during the late pupal or imago stage (e.g. predator abundance, conspecific density, adult food resources) would likely cause ecotype-related differences in mean personality.

Additionally, there was a trend towards lower boldness scores in agricultural butterflies compared to urban conspecifics. This is unexpected because we found in a previous experiment that agricultural males were bolder than woodland and urban conspecifics, independently from their landscape of development (Kaiser et al. 2019a). The two experiments notably differ by their design. As mentioned above, Kaiser et al. (2019a) used a reciprocal transplant design conducted under field conditions, while here we used a common garden approach. Depending on developmental conditions (i.e. laboratory versus outdoor), the ranking of the agricultural ecotype relative to the woodland and urban ecotypes - appears to change, which may indicate that boldness in the Speckled wood is shaped by gene  $\times$  environment interactions (Stamps and Groothuis 2010; Niemelä and Dingemanse 2014). Similarly, sequence affected boldness only as a main effect here (see also Kaiser et al. 2018), while it interacted with the ecotype in the reciprocal transplant experiment (Kaiser et al. 2019a). In both experiments, the sequence effect most likely reflects a habituation process, but outdoor developmental conditions seem to have promoted ecotype-related differences in habituation. Nevertheless, we should remain cautious when interpreting results from these two experiments together at this stage and this hypothesis warrants further study.

Within-individual correlation between the two behavioural traits was low in all studied populations. Contrastingly, we detected a significant among-individual correlation between boldness and activity, but only for butterflies of urban origin. Urban butterflies that were on average bolder were less active in the greenhouse test. Although the direction of this correlation is unexpected given that boldness and activity are

 
 Table 2
 Among- and within-individual correlation coefficients between boldness and activity (with 95% credible intervals), for each ecotype separately. Significant correlation estimates are highlighted in bold

	Among-individual correlation	Within-individual correlation
Woodland	-0.333 [-0.825; 0.398]	-0.200 [-0391; 0.025]
Agricultural	0.181 [-0.597; 0.813]	-0.046 [-0.174; 0.217]
Urban	-0.699 [-0.902; -0.137]	0.029 [-0.145; 0.179]

expected to be positively related (Réale et al. 2010), our results suggest that cities favour different behavioural syndromes compared to rural areas. Interestingly, studies on birds showed that behavioural syndromes tend to break down with urbanization (Scales et al. 2011; Bókony et al. 2012; Carrete and Tella 2017; but see Hardman and Dalesman 2018), while we found that urban butterflies actually have tighter behavioural syndromes. We believe this difference among taxa probably relates to distinct effects of urbanization on birds versus small-sized animals such as butterflies. Cities are highly productive habitats for city-dwelling birds (Shochat et al. 2006) and bird predation rates are lower than in surrounding rural areas (Eötvös et al. 2018). Therefore, at least some of the selective pressures in cities can be assumed to be relaxed for birds. By contrast, urban-heat-island effects (Merckx et al. 2018b; Brans et al. 2018), mechanical disturbances due to green infrastructure management, and high turnover of semi-natural areas (Parris 2016) are important stressors for city invertebrates. In particular, (micro-) climatic alterations and the increased likelihood of drought events, as well as their indirect impacts on food quantity and quality, are expected to be important for ectothermic invertebrates such as *P. aegeria*, which was originally a woodland species and is hence drought-sensitive (Oliver et al. 2015). In line with recent work on rural-urban differentiation in damselflies (Tüzün et al. 2017a), urban areas may thus represent generally more stressful low quality habitats, favouring the emergence of particular behavioural syndromes in city invertebrates. Our sampling design, including families from ten regions, reduces the probability that the observed difference is primarily driven by genetic drift, which may be an important non-adaptive process shaping urban-rural differences (Rivkin et al. 2019). Consequently, future laboratory experiments simulating drought stress (see Talloen et al. 2004), may provide useful insights on the adaptive value of trait integration under stressful conditions.

## Conclusion

Using a butterfly species that has recently expanded its ecological niche, we aimed to unravel the effect of human-altered landscapes on personality traits. While mean levels of boldness (but not activity) were weakly related to the ecotype, covariation between personality traits differed according to the landscape of origin. Although we cannot fully rule out maternal effects, our common garden approach suggests that this difference has at least partly a genetic basis and results from micro-evolutionary changes (Alberti et al. 2017). Currently, about half of the Earth's land surface is covered by anthropogenic habitats (Ellis et al. 2010), some of which – like urban areas – are expected to experience a global expansion on the relatively short term (Seto et al. 2012). Therefore, understanding how novel environmental conditions shape the evolution of successful species is crucial to implement effective mitigation measures to reduce the negative impact of human-altered habitats on biodiversity in general.

Acknowledgements We thank two anonymous reviewers for their constructive comments, and Camille Turlure for help with catching and rearing butterflies. This is publication number BRC 363 of the Biodiversity Research Centre (ELI/ELIB, UCLouvain).

**Funding** AK is a research fellow with the Belgian Fund of Scientific Research F.S.R.-FNRS. The research was supported by PDR grant FNRS PDR T.0188.14 and an ARC research grant of Fédération Wallonie-Bruxelles and UCLouvain (ARC-grant 17/22–086) to HVD.

**Data availability** The data set generated and analysed during the current study is available from the corresponding author on reasonable request.

## **Compliance with ethical standards**

**Conflict of interest** The authors declare that they have no conflict of interest.

**Ethical approval** Our experimental protocol complies with all institutional guidelines at UCLouvain and the F.R.S.-FNRS on invertebrate research. No permit was necessary to perform the experiments described in the study.

## References

- Alberti M, Marzluff J, Hunt VM (2017) Urban driven phenotypic changes: empirical observations and theoretical implications for ecoevolutionary feedback. Philos Trans R Soc B Biol Sci 372: 20160029. https://doi.org/10.1098/rstb.2016.0029
- Allendorf FW, Hard JJ (2009) Human-induced evolution caused by unnatural selection through harvest of wild animals. Proc Natl Acad Sci 106:9987–9994. https://doi.org/10.1073/pnas.0901069106
- Baxter-Gilbert J, Riley JL, Whiting MJ (2019) Bold New World: urbanization promotes an innate behavioral trait in a lizard. Behav Ecol Sociobiol 73:105. https://doi.org/10.1007/s00265-019-2713-9
- Bell AM (2005) Behavioural differences between individuals and two populations of stickleback (Gasterosteus aculeatus). J Evol Biol 18:464–473. https://doi.org/10.1111/j.1420-9101.2004.00817.x
- Bergerot B, Merckx T, Van Dyck H, Baguette M (2012) Habitat fragmentation impacts mobility in a common and widespread woodland butterfly: do sexes respond differently? BMC Ecol 12:5. https://doi. org/10.1186/1472-6785-12-5

- Bókony V, Kulcsár A, Tóth Z, Liker A (2012) Personality traits and behavioral syndromes in differently urbanized populations of house sparrows (*Passer domesticus*). PLoS One 7:e36639. https://doi.org/ 10.1371/journal.pone.0036639
- Brans KI, Engelen JMT, Souffreau C, De Meester L (2018) Urban hottubs: local urbanization has profound effects on average and extreme temperatures in ponds. Landsc Urban Plan 176:22–29. https://doi. org/10.1016/j.landurbplan.2018.03.013
- Carrete M, Tella JL (2017) Behavioral correlations associated with fear of humans differ between rural and urban burrowing owls. Front Ecol Evol 5:54. https://doi.org/10.3389/fevo.2017.00054
- Charmantier A, Demeyrier V, Lambrechts M, Perret S, Grégoire A (2017) Urbanization Is Associated with Divergence in Pace-of-Life in Great Tits. Urbanization is associated with divergence in pace-of-life in great tits Front Ecol Evol 5:5. https://doi.org/10. 3389/fevo.2017.00053
- Concepción ED, Moretti M, Altermatt F, Nobis MP, Obrist MK (2015) Impacts of urbanisation on biodiversity: the role of species mobility, degree of specialisation and spatial scale. Oikos 124:1571–1582. https://doi.org/10.1111/oik.02166
- Diamond SE, Chick LD, Perez A, Strickler SA, Zhao C (2018) Evolution of plasticity in the city: urban acorn ants can better tolerate more rapid increases in environmental temperature. Conserv Physiol 6:1– 12. https://doi.org/10.1093/conphys/coy030
- Dingemanse NJ, Dochtermann NA (2013) Quantifying individual variation in behaviour: mixed-effect modelling approaches. J Anim Ecol 82:39–54. https://doi.org/10.1111/1365-2656.12013
- Dingemanse NJ, Dochtermann NA, Nakagawa S (2012) Defining behavioural syndromes and the role of 'syndrome deviation' in understanding their evolution. Behav Ecol Sociobiol 66:1543–1548. https://doi.org/10.1007/s00265-012-1416-2
- Dingemanse NJ, Wright J, Kazem AJN et al (2007) Behavioural syndromes differ predictably between 12 populations of three-spined stickleback. J Anim Ecol 76:1128–1138. https://doi.org/10.1111/j. 1365-2656.2007.01284.x
- Ducatez S, Humeau A, Congretel M, Fréville H, Baguette M (2014) Butterfly species differing in mobility show different structures of dispersal-related syndromes in the same fragmented landscape. Ecography (Cop) 37:378–389. https://doi.org/10.1111/j.1600-0587.2013.00365.x
- Ellis EC, Goldewijk KK, Siebert S et al (2010) Anthropogenic transformation of the biomes, 1700 to 2000. Glob Ecol Biogeogr 19:589– 606. https://doi.org/10.1111/j.1466-8238.2010.00540.x
- Eötvös CB, Magura T, Lövei GL (2018) A meta-analysis indicates reduced predation pressure with increasing urbanization. Landsc Urban Plan 180:54–59. https://doi.org/10.1016/j.landurbplan.2018. 08.010
- Evans J, Boudreau K, Hyman J (2010) Behavioural syndromes in urban and rural populations of song sparrows. Ethology 116:588–595. https://doi.org/10.1111/j.1439-0310.2010.01771.x
- Evans KL, Gaston KJ, Frantz AC, Simeoni M, Sharp SP, McGowan A, Dawson DA, Walasz K, Partecke J, Burke T, Hatchwell BJ (2009) Independent colonization of multiple urban centres by a formerly forest specialist bird species. Proc R Soc B Biol Sci 276:2403–2410. https://doi.org/10.1098/rspb.2008.1712
- Hadfield JD (2010) MCMC methods for multi-response generalized linear mixed models: the MCMCglmm R package. J stat Softw 33:1– 22. https://doi.org/10.18637/jss.v033.i02
- Hardman SI, Dalesman S (2018) Repeatability and degree of territorial aggression differs among urban and rural great tits (*Parus major*). Sci Rep 8:5042. https://doi.org/10.1038/s41598-018-23463-7
- Hobbs RJ, Higgs E, Harris JA (2009) Novel ecosystems: implications for conservation and restoration. Trends Ecol Evol 24:599–605. https:// doi.org/10.1016/j.tree.2009.05.012
- Kaiser A, Merckx T, Van Dyck H (2016) The urban Heat Island and its spatial scale dependent impact on survival and development in

butterflies of different thermal sensitivity. Ecol Evol 6:4129–4140. https://doi.org/10.1002/ece3.2166

- Kaiser A, Merckx T, Van Dyck H (2018) Urbanisation and sex affect the consistency of butterfly personality across metamorphosis. Behav Ecol Sociobiol 72:188. https://doi.org/10.1007/s00265-018-2616-1
- Kaiser A, Merckx T, Van Dyck H (2019a) Behavioural repeatability is affected by early developmental conditions in a butterfly. Anim Behav 157:219–226. https://doi.org/10.1016/j.anbehav.2019.08. 006
- Kaiser A, Merckx T, Van Dyck H (2019b) Personality traits influence contest outcome, and vice versa, in a territorial butterfly. Sci Rep 9: 2778. https://doi.org/10.1038/s41598-019-39155-9
- Kern EMA, Langerhans RB (2018) Urbanization drives contemporary evolution in stream fish. Glob Chang Biol 24:3791–3803. https:// doi.org/10.1111/gcb.14115
- Koolhaas J, Korte S, De Boer S et al (1999) Coping styles in animals: current status in behavior and stress-physiology. Neurosci Biobehav Rev 23:925–935. https://doi.org/10.1016/S0149-7634(99)00026-3
- Lapiedra O, Chejanovski Z, Kolbe JJ (2017) Urbanization and biological invasion shape animal personalities. Glob Chang Biol 23:592–603. https://doi.org/10.1111/gcb.13395
- Luniak M, Mulsow R, Walasz K (1990) Urbanization of the European blackbird – expansion and adaptations of urban population. In: Luniak M (ed) Urban ecological studies in central and Eastern Europe. International Symposium Warsaw, Poland, pp 187–200
- McKinney ML (2008) Effects of urbanization on species richness: a review of plants and animals. Urban Ecosyst 11:161–176. https:// doi.org/10.1007/s11252-007-0045-4
- Merckx T, Kaiser A, Van Dyck H (2018a) Increased body size along urbanization gradients at both community and intraspecific level in macro-moths. Glob Chang Biol 24:3837–3848. https://doi.org/10. 1111/gcb.14151
- Merckx T, Souffreau C, Kaiser A, Baardsen LF, Backeljau T, Bonte D, Brans KI, Cours M, Dahirel M, Debortoli N, de Wolf K, Engelen JMT, Fontaneto D, Gianuca AT, Govaert L, Hendrickx F, Higuti J, Lens L, Martens K, Matheve H, Matthysen E, Piano E, Sablon R, Schön I, van Doninck K, de Meester L, van Dyck H (2018b) Bodysize shifts in aquatic and terrestrial urban communities. Nature 558: 113–116. https://doi.org/10.1038/s41586-018-0140-0
- Merckx T, Van Dyck H (2005) Mate location behaviour of the butterfly Pararge aegeria in woodland and fragmented landscapes. Anim Behav 70:411–416. https://doi.org/10.1016/j.anbehav.2004.12.005
- Merckx T, Van Dyck H (2007) Habitat fragmentation affects habitatfinding ability of the speckled wood butterfly, *Pararge aegeria* L. Anim Behav 74:1029–1037. https://doi.org/10.1016/j.anbehav. 2006.12.020
- Miranda AC, Schielzeth H, Sonntag T, Partecke J (2013) Urbanization and its effects on personality traits: a result of microevolution or phenotypic plasticity? Glob Chang Biol 19:2634–2644. https://doi. org/10.1111/gcb.12258
- Nakagawa S, Schielzeth H (2010) Repeatability for Gaussian and non-Gaussian data: a practical guide for biologists. Biol Rev 85:935– 956. https://doi.org/10.1111/j.1469-185X.2010.00141.x
- Niemelä PT, Dingemanse NJ (2014) Artificial environments and the study of 'adaptive' personalities. Trends Ecol Evol 29:245–247. https://doi.org/10.1016/j.tree.2014.02.007
- Oliver TH, Marshall HH, Morecroft MD, Brereton T, Prudhomme C, Huntingford C (2015) Interacting effects of climate change and habitat fragmentation on drought-sensitive butterflies. Nat Clim Chang 5:941–945. https://doi.org/10.1038/nclimate2746
- Parris KM (2016) Ecology of urban environments. Wiley-Blackwell, Chichester, UK
- Piano E, Souffreau C, Merckx T, Baardsen LF, Backeljau T, Bonte D, Brans KI, Cours M, Dahirel M, Debortoli N, Decaestecker E, de Wolf K, Engelen JMT, Fontaneto D, Gianuca AT, Govaert L, Hanashiro FTT, Higuti J, Lens L, Martens K, Matheve H,

Behav Ecol Sociobiol (2020) 74:86

Matthysen E, Pinseel E, Sablon R, Schön I, Stoks R, van Doninck K, van Dyck H, Vanormelingen P, van Wichelen J, Vyverman W, de Meester L, Hendrickx F (2020) Urbanization drives cross-taxon declines in abundance and diversity at multiple spatial scales. Glob Chang Biol 26:1196–1211. https://doi.org/10.1111/gcb.14934

- R Core Team (2020) R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL: https://www.R-project.org/
- Ramírez-Restrepo L, MacGregor-Fors I (2017) Butterflies in the city: a review of urban diurnal Lepidoptera. Urban Ecosyst 20:171–182. https://doi.org/10.1007/s11252-016-0579-4
- Réale D, Garant D, Humphries MM, Bergeron P, Careau V, Montiglio PO (2010) Personality and the emergence of the pace-of-life syndrome concept at the population level. Philos Trans R Soc B Biol Sci 365:4051–4063. https://doi.org/10.1098/rstb.2010.0208
- Réale D, Reader SM, Sol D, McDougall PT, Dingemanse NJ (2007) Integrating animal temperament within ecology and evolution. Biol Rev 82:291–318. https://doi.org/10.1111/j.1469-185X.2007. 00010.x
- Rivkin LR, Santangelo JS, Alberti M, Aronson MFJ, de Keyzer CW, Diamond SE, Fortin MJ, Frazee LJ, Gorton AJ, Hendry AP, Liu Y, Losos JB, MacIvor JS, Martin RA, McDonnell MJ, Miles LS, Munshi-South J, Ness RW, Newman AEM, Stothart MR, Theodorou P, Thompson KA, Verrelli BC, Whitehead A, Winchell KM, Johnson MTJ (2019) A roadmap for urban evolutionary ecology. Evol Appl 12:384–398. https://doi.org/10.1111/eva. 12734
- Royauté R, Buddle CM, Vincent C (2015) Under the influence: sublethal exposure to an insecticide affects personality expression in a jumping spider. Funct Ecol 29:962–970. https://doi.org/10.1111/ 1365-2435.12413
- Scales J, Hyman J, Hughes M (2011) Behavioral syndromes break down in urban song sparrow populations. Ethology 117:887–895. https:// doi.org/10.1111/j.1439-0310.2011.01943.x
- Schuett W, Delfs B, Haller R, Kruber S, Roolfs S, Timm D, Willmann M, Drees C (2018) Ground beetles in city forests: does urbanization predict a personality trait? PeerJ 6:e4360. https://doi.org/10.7717/ peerj.4360
- Senar JC, Garamszegi LZ, Tilgar V, Biard C, Moreno-Rueda G, Salmón P, Rivas JM, Sprau P, Dingemanse NJ, Charmantier A, Demeyrier V, Navalpotro H, Isaksson C (2017) Urban great tits (*Parus major*) show higher distress calling and pecking rates than rural birds across Europe. Front Ecol Evol 5. https://doi.org/10.3389/fevo.2017.00163
- Sepp T, McGraw KJ, Kaasik A, Giraudeau M (2018) A review of urban impacts on avian life-history evolution: does city living lead to slower pace of life? Glob Chang Biol 24:1452–1469. https://doi. org/10.1111/gcb.13969
- Serruys M, Van Dyck H (2014) Development, survival, and phenotypic plasticity in anthropogenic landscapes: trade-offs between offspring quantity and quality in the nettle-feeding peacock butterfly. Oecologia 176:379–387. https://doi.org/10.1007/s00442-014-3016-5
- Seto KC, Guneralp B, Hutyra LR (2012) Global forecasts of urban expansion to 2030 and direct impacts on biodiversity and carbon pools. Proc Natl Acad Sci 109:16083–16088. https://doi.org/10.1073/pnas. 1211658109

- Settele J, Kudrna O, Harpke A et al (2008) Climatic Risk Atlas of European Butterflies. In: Climatic risk atlas of European butterflies. Pensoft, Sofia-Moscow
- Shochat E, Warren P, Faeth S et al (2006) From patterns to emerging processes in mechanistic urban ecology. Trends Ecol Evol 21:186– 191. https://doi.org/10.1016/j.tree.2005.11.019
- Shreeve TG (1986) Egg-laying by the speckled wood butterfly (*Pararge aegeria*): the role of female behaviour, host plant abundance and temperature. Ecol Entomol 11:229–236. https://doi.org/10.1111/j. 1365-2311.1986.tb00298.x
- Shreeve TG (1984) Habitat selection, mate location, and microclimatic constraints on the activity of the speckled wood butterfly *Pararge aegeria*. Oikos 42:371–377. https://doi.org/10.2307/3544407
- Sih A, Bell A, Johnson JC (2004) Behavioral syndromes: an ecological and evolutionary overview. Trends Ecol Evol 19:372–378. https:// doi.org/10.1016/j.tree.2004.04.009
- Sol D, Lapiedra O, González-Lagos C (2013) Behavioural adjustments for a life in the city. Anim Behav 85:1101–1112. https://doi.org/10. 1016/j.anbehav.2013.01.023
- Sprau P, Mouchet A, Dingemanse NJ (2017) Multidimensional environmental predictors of variation in avian forest and city life histories. Behav Ecol 28:59–68. https://doi.org/10.1093/beheco/arw130
- Stamps J, Groothuis TGG (2010) The development of animal personality: relevance, concepts and perspectives. Biol Rev 85:301–325. https:// doi.org/10.1111/j.1469-185X.2009.00103.x
- Sullivan AP, Bird DW, Perry GH (2017) Human behaviour as a longterm ecological driver of non-human evolution. Nat Ecol Evol 1: 0065. https://doi.org/10.1038/s41559-016-0065
- Talloen W, Van Dyck H, Lens L (2004) The cost of melanization: butterfly wing coloration under environmental stress. Evolution 58: 360–366. https://doi.org/10.1111/j.0014-3820.2004.tb01651.x
- Tüzün N, Müller S, Koch K, Stoks R (2017a) Pesticide-induced changes in personality depend on the urbanization level. Anim Behav 134: 45–55. https://doi.org/10.1016/j.anbehav.2017.10.007
- Tüzün N, Op de Beeck L, Brans KI et al (2017b) Microgeographic differentiation in thermal performance curves between rural and urban populations of an aquatic insect. Evol Appl 10:1067–1075. https:// doi.org/10.1111/eva.12512
- Urszán TJ, Garamszegi LZ, Nagy G, Hettyey A, Török J, Herczeg G (2015) No personality without experience? A test on *Rana dalmatina* tadpoles. Ecol Evol 5:5847–5856. https://doi.org/10. 1002/ece3.1804
- Van Dyck H, Matthysen E (1998) Thermoregulatory differences between phenotypes in the speckled wood butterfly: hot perchers and cold patrollers? Oecologia 114:326–334. https://doi.org/10.1007/ s004420050454
- Verbeek MEM, Boon A, Drent PJ (1996) Exploration, aggressive behaviour and dominance in pair-wise confrontations of juvenile male great tits. Behaviour 133:945–963. https://doi.org/10.1163/ 156853996X00314
- Winchell KM, Reynolds RG, Prado-Irwin SR, Puente-Rolón AR, Revell LJ (2016) Phenotypic shifts in urban areas in the tropical lizard *Anolis cristatellus*. Evolution 70:1009–1022. https://doi.org/10. 1111/evo.12925

Publisher's note Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.